**Original scientific paper** 

# DRONE SELECTION FOR FOREST SURVEILLANCE AND FIRE DETECTION USING INTERVAL VALUED NEUTROSOPHIC EDAS METHOD

## Alize Yaprak Gul<sup>1</sup>, Emre Cakmak<sup>2</sup>, Atiye Ece Karakas<sup>3</sup>

<sup>1</sup>Department of Industrial Engineering, Istanbul Technical University, Istanbul, Turkey <sup>2</sup>Department of Industrial Engineering, Istinye University, Istanbul, Turkey <sup>3</sup>Department of Industrial Engineering, Piri Reis University, Istanbul, Turkey

Abstract. Forest fires are one of the major causes for deforestation resulting in significant economic and environmental losses. The application of drones has been extended to various areas including disaster management. Since drones offer numerous advantages like real-time surveillance, task planning capabilities and autonomy, they are utilized in early detection systems for forest fires. The selection of a drone type for this purpose involves a complex system of multiple factors and conflicting information, for which the use of multi-criteria decision-making (MCDM) methods have been found to be yielding effective results. The aim of this study is to present a decision framework for drone selection problem in the context of forest fire surveillance and detection. This study contributes by (i) pointing out to the gap that the drone selection problem for forest surveillance and fire detection has been sparsely addressed, (ii) presenting an extensive literature review, (iii) extracting the relevant criteria through a literature review and interviews with the experts in field, (iv) assessing the alternatives by the proposed framework based on interval valued neutrosophic evaluation based on distance from average solution (IVN EDAS) method. The proposed framework is demonstrated by a case study consisting of four drone alternatives and 14 criteria. In accordance with the extant literature, the criteria related to the visual capabilities and diagnosis are evaluated as the most crucial features. A sensitivity analysis is carried out to check for the robustness by varying the criteria weights and a comparative analysis is conducted with interval valued neutrosophic technique for preference by similarity to the ideal solution (IVN TOPSIS) and interval valued neutrosophic combinative distance-based assessment (IVN CODAS) methods to validate the veracity of the method.

Key words: Drone Selection, Forest Surveillance, Fire Detection, Interval Valued Neutrosophic Sets, MCDM, EDAS

Received: October 28, 2023 / Accepted February 27, 2024

Corresponding author: Alize Yaprak Gul

Department of Industrial Engineering, Istanbul Technical University, 34367 Macka, Besiktas, Istanbul, Turkey E-mail: gul22@itu.edu.tr

<sup>© 2024</sup> by University of Niš, Serbia | Creative Commons License: CC BY-NC-ND

#### 1. INTRODUCTION

Forests are vital for the sustainability of the ecosystem and human life. They play a crucial role in counteracting climate change and its worsening effects. Among the deforestation causes, wildfires have been responsible for about 34% of the tree cover loss since 2019 [1]. Increasing wildfires and deforestation exacerbate the effects of climate change which lead to a vicious cycle of drier seasons followed by more frequent and intense fires. One of the most devastating wildfires in history occurred in the 2019-2020 Australian bushfire season, the burnt area had been estimated to be between 24.3 and 33.8 million hectares [2], which was effective from June 2019 to May 2020. In total, 33 people have lost lives, 3.094 houses have been destroyed and the direct economic cost has been estimated to be \$2.5 billion in addition to \$4-5 billion worth of losses on the Australian food and agriculture industry [3]. Deforestation substantially damages the economy and the ecosystem, which is avoidable to some extent. One of the ways to reduce these damages is to take environmental precautions and extinguish the fire as soon as they occur. As time passes, the destructive effect of wildfires magnifies abruptly. A rule has been mentioned i.e., the amount of water that is required to suppress a fire burning for a minute is multiplied by ten when the burning time doubles [4]. Thus, early detection and taking countermeasures on time are crucial to prevent significant damage.

For this reason, various technologies have found its application for monitoring and detection of forest fire. The technologies used can be summarized as satellite-based, sensorbased, camera-based, mobile biological sensing, unmanned aerial vehicle (UAV), radioacoustic based, neural network based and fuzzy logic based technologies according to the recent classification done by Chowdary et al. [5]. In practice, many of these technologies cannot be deployed as effective stand-alone systems, instead a combination of these technologies is utilized to enhance the accuracy and efficiency in detection. However, each of these technologies are associated with certain advantages and disadvantages. In comparison to the satellite-based and ground-based technologies with static topology, the use of drones brings several useful properties such greater maneuverability and real-time surveillance, especially in the areas that are difficult to access. Further, drones provide autonomy, task planning capabilities and self-healing properties [6]. The radical growth in the sensor and micro-processing technologies have facilitated their usage in different areas including the fire detection systems. The sensor technologies serve with improved preciseness and fewer false alarms notably when utilizing different types of sensors. Thus, the application of drones integrated with different technologies as sensors and neural networks can realize an effective and self-sufficient solution for the surveillance and detection of forest fires. The application of drones and its potential for the respective problem requires more attention for the future research.

There are several factors affecting the performance of the deployed drone type in the respective context. Thus, the problem of its selection contains several ambiguities, that arise from several sources such as multiple evaluation attributes, the uncertainty and bias incorporated into the human decision-making process as well as the factors and the characteristics affecting the performance of a drone. According to Hristozov and Zlateva [7], the performance of a drone is influenced by its internal attributes, which are the technical features, and the external attributes such as the characteristics of the area, weather and other environmental conditions. Some of these attributes may be in trade-off or have interrelationships with one and other. Such complex decision-making problems that

involve inexplicable and unquantifiable relationships are broadly handled by the multicriteria decision-making (MCDM) methods, since they yield appropriate results with significantly lower computation time and complexity.

The vagueness and uncertainty present in information can be successfully represented by fuzzy logic introduced by Zadeh [8]. Turksen [9] extended fuzzy sets to interval valued fuzzy sets to enable for a range of values to be assigned in the membership information. Smarandache [10] introduced the neutrosophic sets, which enable the representation of hesitancy in addition to the membership and non-membership information, as an extension to intuitionistic fuzzy sets [11]. Moreover, fuzzy MCDM methods are particularly helpful in addressing the decision-making problems with uncertainty and ambiguity resulting from the subjective evaluations and vague relationships among criteria and alternatives. Among many proposed MCDM methods, evaluation based on distance from average solution (EDAS) method by Keshavarz Ghorabaee et al. [12] processes the conflicting criteria and performs better at considering the intangibility and vagueness in the decision-making process [13]. It also handles with the biased information better since its calculation is based on the assessment by the average solution. For this reason, this study aims to handle the drone selection problem for monitoring and detection of forest fires by proposing a framework using interval valued neutrosophic (IVN) sets to present the vague and uncertain linguistic information and EDAS method to evaluate the alternatives based on the conflicting criteria. The findings from the case study demonstrate that "the camera accuracy" is the most significant feature for a drone used in forest surveillance and fire detection. The "ingress protection rating", "maximum flight time" are the second and third most significant characteristics. The rest of the criteria are ranked with the following order: "wind resistance", "camera resolution", "zoom camera", "maximum operation altitude", "maximum horizontal speed", "maximum hover time", "operation frequency", "obstacle sensors", "maximum takeoff weight", "charging time" and "hovering accuracy".

This study contributes to literature by pointing out to the gap that the drone selection problem for forest surveillance and fire detection has been sparsely addressed, by presenting an extensive literature review on the technologies deployed in forest monitoring and fire detection, and by extracting the relevant criteria through an extensive literature review and interviews with the experts in field. As a result, four drone alternatives and 14 criteria have been identified. The alternatives are then evaluated based on the proposed framework using IVN EDAS method. Moreover, a sensitivity analysis is conducted in order to check for the robustness of the results by varying the criteria weights. The veracity of the results is validated by carrying out a comparative analysis with interval valued neutrosophic technique for preference by similarity to the ideal solution (IVN TOPSIS) method and interval valued neutrosophic combinative distance-based assessment (IVN CODAS) method.

The rest of this paper is structured as follows. Section 2 presents an in-depth literature review on the other technologies and the application of drones in the respective context along with the examination of drone selection problem by the MCDM methods. Section 3 provides the methodology of IVN EDAS method with the preliminaries of interval valued neutrosophic sets. Then, Section 4 presents the application of the method on a case study that is concluded with a sensitivity analysis and a comparative analysis with IVN TOPSIS and IVN CODAS methods. Lastly, Section 5 gives the conclusion with the recommendations for future study.

### 2. LITERATURE REVIEW

Monitoring and early detection of a possible fire are as crucial as the fast response and suppression of wildfires to avoid irreversible damages. Several studies have examined the technologies used in forest fires and investigated their benefits and drawbacks. We first summarize these technologies and then discuss the advantages of the application of drones in forest surveillance and fire detection. Den Breejen et al. [14] categorized the techniques used in forest fire monitoring and detection into three groups as ground-based systems, manned aerial vehicle-based systems, and satellite-based systems. In their classification, a ground-based system monitors and gathers information by utilizing equipment with static topology. Manned aerial vehicle-based systems describe systems that use an aerial vehicle controlled by a human operator to patrol the area for the detection of a possible fire. In satellite-based systems, the imagery taken through the satellite is collected and processed with the image-processing techniques. Recent technological advancements in the field of wireless networks, microprocessors, image-processing and artificial intelligence have enabled a variety and a combination of these methods for the monitoring and detection purposes of forest fires. Alkhatib [4] generalized the detection and monitoring systems as suppression and detection techniques used by the authorities, satellite-based systems, optical sensors, and digital camera and wireless sensor networks (WSN). The suppression and detection techniques used by the authorities include watch towers, water tankers, lightning detectors, etc. Among the optical sensor and digital camera technologies, the video-cameras sensitive to smoke, infrared thermal imaging cameras and light detection and ranging systems (LIDAR) are widely applied. In WSNs, the area of interest is surveilled by multiple wireless nodes that are equipped with different types of sensors and microprocessors. Depending on the movement capability of nodes, WSNs are classified as WSN with static topology and WSN with dynamic topology. In WSNs with dynamic topology, the nodes follow a common protocol. Chowdary et al. [5] extended this classification by sensor-based, neural networks-based, drone/airborne-based, fuzzy logic, mobile biological sensing-based, and radio acoustic-based techniques. By fuzzy logic techniques, a fuzzy logic algorithm is created and applied to the visual data gathered from various sources. The proposed fuzzy logic algorithm may use instances such as temperature, smoke, light, humidity, and distance [15] and shape, size, and motion variation of the fire [16] for which membership functions are generated. Then for the verification, the performance of the decision-making procedure is evaluated either by simulations or real fire datasets. In mobile biological sensing-based detection, certain animal groups may be equipped with sensor and global positioning system (GPS) devices transmitting data and location information, which are examined for sudden environmental changes such as temperature or humidity or changes in animal behavior. In practice, many systems combine multiple techniques together for more precision in monitoring and detection. Many forest fire surveillance systems aim to deploy a self-sufficient system that require little or no maintenance or supervision. Ideally, the deployed system should monitor the forest and detect the fire as soon as possible with a low false alarm chance. Also, it should inform the authorities automatically. From the studies that have dealt with these technologies' advantages and disadvantages, it has been found that the satellite-based techniques have a low temporal and spatial resolution [17]. The quality of imagery is highly affected by the terrain, time of day, and weather [5] resulting in limited coverage, limited precision, and a lack of real-time data reporting [6]. Radio acoustic-based methods are

prone to fire localization errors and a high chance of false alarms [5]. In camera-based technologies, immobile cameras equipped with multiple types of sensors are positioned at specified locations for surveillance. Immobile cameras provide a limited surveillance range [18] and image and video processing techniques are associated with significant preprocessing time and effort [5]. Generally used miniaturized infrared cameras have low sensitivity, which leads to a high false alarm possibility [5, 19]. Neural network-based techniques require large datasets and heavy model sizes, which are also prone to various false alarms [5]. Sensor-based techniques provide more accuracy in the surveillance and detection of forest fires than satellite-based techniques as these are deployable in areas that are not observable by satellites [20]. However, WSNs with static topology require regular maintenance of the infrastructure and they have limited coverage and effectiveness [21]. As these systems operate as static infrastructure, they are prone to get destroyed in the event of a fire which may cause additional replacement costs. Their effectiveness in terms of coverage and resolution is directly determined by the investment made into the deployed system [6].

Satellite-based and ground-based techniques such as camera surveillance and WSNs with static topology may be insufficient for precise real-time monitoring of large areas. Drones offer greater maneuverability and real-time surveillance capabilities in areas hard to access. Recent progressive improvements in micro-processing, imaging, and drone technologies facilitate lower costs and wider usage of these technologies. The type of sensors utilized in the surveillance drones can be chosen depending on the type of possible fire and topology of the area to increase precision in detection. In the equipment with different types of sensors, a similarity index of measurements helps to mitigate the chance of false alarms [22]. In addition, the major conveniences of drones are that they provide great autonomy and task-planning capabilities with efficient allocation, self-organization, and self-healing properties [6].

For the sustainability and security of the system, the selection of the drone model is as important as the system architecture, sensors, and its other fundamental components. Among a great variety of drones available on the market, a drone needs to be selected that is most appropriate to the characteristics of the area and can satisfy the necessary requirements for the tasks.

Many factors impact the operational performance of a drone and thus its selection, these factors can be viewed as external factors and internal characteristics or attributes of a drone [7]. The external factors are the disaster type, the characteristics of the area of interest, and the weather and other environmental conditions. For instance, it has been found that strong wind and precipitation affect drones' performance significantly [23, 24]. Under the influence of the strong wind, the movement capability of a drone is significantly disrupted which may result in the deviation from its direction and poor stabilization during hovering. The stabilization of a drone is especially important for the quality of imaging. Other environmental conditions such as temperature and altitude of the area must be considered since drones are not able to operate at all conditions.

With the wide-ranging features and integrable nature of drones, they have a substantial role as a dynamic, deployable, and controllable component of the Internet of Things infrastructure [25, 26]. Drones have been manufactured and used primarily for military and security purposes. Their application areas are extended to natural disasters, agricultural fields and smart farming, cargo handling, delivery of medical goods, traffic management, conservation, and land monitoring, climate change and urban planning, and more.

Therefore, they are available with various features and attributes such as drone type, weight, speed, takeoff weight, hover time, charging time, flight time, operating altitude, etc. [27, 28]. Yet some of these attributes are in trade-off with one another. As an example, drones with higher speed capability have lower payload capacity [29, 30], and typically drones with slower speed can take shorter distances and handle higher payloads [31]. As a result, the drone selection problem must consider these multiple external factors and conflicting internal attributes, which makes it a highly complex and a challenging task.

This study focuses on the drone selection problem for monitoring and detection of forest fires. The application of drones in the wildfire context is studied by many authors. To our best knowledge, the studies on the drone usage in wildfires are mainly gathered on three scopes: vision-based techniques and (deep learning) image-processing techniques [19, 32, 33], the system architecture with communication, data receiving, and GIS modules [34, 35], and the area coverage and coordination of multiple drones [5, 36, 37, 38]. Yuan et al. [19] presented the first thorough analysis of technologies that use drones for fire detection, diagnosis, prognosis, image vibration elimination, and cooperative control of drones. Bailon-Ruiz and Lacroix [39] reviewed a selection of studies that examine drone application in wildfire remote sensing based on an autonomy perspective using three metrics: situation awareness, decisional ability, and collaboration ability. Akhloufi et al. [6] presented an extensive review with a focus on onboard sensor instruments, fire perception algorithms, and coordination strategies. The prior studies reviewed by [6, 19, 39] assumed that a set of drones are available and suited for their specific use and thus the drone selection problem has not been mentioned. Regarding the drone selection problem, Hristozov and Zlateva [7] introduced a performance mapping model as a concept model for drone selection in specific disaster conditions to aid the drone selection decision process. To best of our knowledge, in the literature only the study by Pamučar et al. [40] dealt with the drone selection problem related to forest fire uses. Their study focused on determining a drone model that can be used as a physical fire suppression tool that is controlled by operators in a ground central system. Our study distinguishes from the prior study in that it deals with the drone selection problem for the monitoring and detection of forest fires for its application as an early warning system that requires minimum human assistance. Using a drone for monitoring and detection purposes and as a fire suppression tool require different technical characteristics. A network of drones can act as mobile wireless sensor network, where the drones are equipped with various sensors to collect data on temperature, smoke and other parameters. The collected data should be transmitted to a head cluster thereafter it is transmitted to gateway to get processed. The drones in a WSN require to follow a protocol for self-organized deployment and coverage of the area. The energy efficiency in WSN communication is of major concern for an effective communication [41, 42]. Likewise, each drone has a limited payload. The higher the payload of a drone, the higher is its battery consumption and shorter is its flight time. Therefore, the weight of a drone should be ideally kept low [43]. Thus, the energy consumption of a drone and of the WSN communication are two fundamental issues concerning the effectiveness of the whole as a fire detection system, one contributing to the movement capability and other to the efficient data transmission. To utilize a drone additionally as a firefighting tool, it should be equipped with fire extinguishing instruments. As a result, it would increase its payload and shorten the endurance of a drone. Therefore, we believe that the drone selection for monitoring and detection and for firefighting purpose should be examined separately. Ideally, both drone networks shall be

integrated to each other as first one would act as an early warning system signaling the latter one to counteract the starting fire timely.

To address the drone selection problem, neutrosophic sets are preferred to represent the ambiguity present in the linguistic assessments of the decision makers (DMs). Different types of uncertainties are associated in the decision-making processes, which can be considered with the help of fuzzy logic introduced by Zadeh [8]. Fuzzy sets have been extended to intuitionistic fuzzy sets by Atanassov [11] which describe both the membership and the non-membership degree. To enable the representation of indeterminacy in information of which the intuitionistic fuzzy sets lack, neutrosophic sets have been developed by Smarandache [10]. Following that, different extensions of neutrosophic sets, have been introduced such as interval-valued neutrosophic sets, type-2 neutrosophic sets, 2-tuple linguistic neutrosophic sets. Numerous studies have applied EDAS method in neutrosophic environment of these extensions such as single-valued neutrosophic sets [44], interval valued neutrosophic sets [45], type-2 neutrosophic sets [46] and bipolar neutrosophic sets [47].

The MCDM methods have found a great variety of application areas thanks to their beneficial aspects. In the presence of complex systems which incorporate multiple criteria and when the interrelationships among the systems' elements are vague and inexplicable by the mathematical models, MCDM methods provide a structured (group) decision making process that is easy to implement, handles the uncertainty, can deal with both quantitative and qualitative data, and provides fast and relatively reliable solutions. It is basically used to select or prioritize alternatives based by assessing on a set of criteria or attributes. MCDM methods are used in various types of decision making problems such as the selection of suppliers [48], the evaluation of potential locations for specific uses [49, 50], but also is applied for the strategy selection [51, 52], performance evaluation [53, 54], and risk evaluation [55] among many others. The EDAS method is utilized by various recent studies with different extensions and application areas. Simic et al. [46] used a type-2 neutrosophic number based threshold-based attribute ratio analysis (ITARA) method integrated with EDAS model for route selection of petroleum transportation. Menekse et al. [56] applied a Pythagorean fuzzy criteria importance through intercriteria correlation (CRITIC) method integrated with EDAS model for the selection of an additive manufacturing process for automotive industry. Dhumras and Bajaj [57] used a picture fuzzy soft Dombi EDAS model for strategy selection for sustainable and smart (robotic) agrifarming.

There are several studies that dealt with the drone selection problem using MCDM methods in other application areas of drones. To give a few examples from the recent studies, for the military purposes a fuzzy weighted average algorithm is applied for group decision-making by Lin and Hung [58]. Hamurcu and Eren [59] proposed a methodology based on analytical hierarchy process (AHP) and technique for preference by similarity to the ideal solution (TOPSIS), and a fuzzy analytical hierarchy process-vise kriterijumska optimizacija i kompromisno resenje (AHP-VIKOR) hybrid model is proposed by Radovanović et al. [60]. The AHP-TOPSIS method is used by several studies for the civilian application areas [61, 62]. With the radical growth in the E-commerce, delivery by drones has gained significant attention as a delivery tool since delivery in conventional ways has become rather inconvenient due to high delivery times and increasing costs in transportation. In this context, Nur et al. [28] performed an interval valued inferential fuzzy (IVIF) TOPSIS method to find the most suitable drone for package delivery purposes in

urban and rural areas and Banik et al. [63] proposed a graph theory and matrix approach to select a drone for the delivery of medical supplies. The examination of the literature shows that the drone selection for the usage against forest fires has been addressed only by Pamučar et al. [40] in terms of selecting a drone with the capability of extinguishment. No prior study has focused on the drone selection solely for the purpose of forest surveillance and fire detection. The next section presents the proposed methodology.

#### 3. METHODOLOGY

The EDAS method, introduced by Keshavarz Ghorabaee et al. [12], is a distance-based method similar to TOPSIS method and VIKOR method. Instead of using positive ideal and negative ideal solutions, EDAS method evaluates the alternatives by using the positive and negative distance to the average solution. This aspect reduces the impact of DMs' bias. It recognizes the conflicting criteria [12, 64] and represents the ambiguity in the decision-making process [13]. The presence of the conflicting criteria makes EDAS method more suitable for this problem. However, the MCDM methods are prone to the rank reversal phenomenon [65] and the results should be further checked against this issue. It occurs when the introduction or the exclusion of a non-optimal alternative modifies the optimal and/or the least preferred alternative. It has been found that the EDAS method performs better than TOPSIS in terms of robustness to the rank reversal phenomenon [66].

The interval valued neutrosophic sets are used to represent the linguistic evaluations of the DMs since this set allows for an enriched representation of information since only the lower values of the parameters (membership, hesitancy, and non-membership) must be equal to or lower than 3, and the sum of upper values can exceed 3 allowing for a flexible information representation. Therefore, the EDAS method under the interval valued neutrosophic environment is selected as an appropriate method to consider the ambiguity in decision making and conflicting criteria. In this section, we introduce the IVN EDAS method and provide the details about its implementation.

#### 3.1 Preliminaries for Interval Valued Neutrosophic Sets

**Definition 1:** Let *X* be the universe. For each element *x* in *X*, an IVN set is be defined by three parameters; the truth-membership  $T_N(x)$ , indeterminacy (hesitancy)-membership  $I_N(x)$ , and the falsity-membership  $F_N(x)$ , where  $T_N = \begin{bmatrix} T & U \\ N(x) & T & U \\ N(x) & \Xi \end{bmatrix}$ ,  $I_N(x) = \begin{bmatrix} I & U \\ N(x) & T & U \\ N(x) & \Xi \end{bmatrix}$ , and  $F_N(x) = \begin{bmatrix} F & U \\ N(x) & F & U \\ N(x) & \Xi \end{bmatrix}$ .

An interval valued neutrosophic number (IVNN) satisfies the following condition that the sum of the lower values of these three elements must be lower than or equal to 3:  $0 \le T_{N(x)}^{L} + I_{N(x)}^{L} + F_{N(x)}^{L} \le 3$ . Thus, an IVN set is defined as in Eq. (1) [67]:

$$\mathbf{N} = \left\{ \left\langle \mathbf{x}, \left[ \mathbf{T}_{\mathbf{N}(\mathbf{x})}^{\mathrm{L}}, \mathbf{T}_{\mathbf{N}(\mathbf{x})}^{\mathrm{U}} \right], \left[ \mathbf{I}_{\mathbf{N}(\mathbf{x})}^{\mathrm{L}}, \mathbf{I}_{\mathbf{N}(\mathbf{x})}^{\mathrm{U}} \right], \left[ \mathbf{F}_{\mathbf{N}(\mathbf{x})}^{\mathrm{L}}, \mathbf{F}_{\mathbf{N}(\mathbf{x})}^{\mathrm{U}} \right] \right\} | \mathbf{x} \in \mathbf{X} \right\}.$$
(1)

Let *a* and *b* be two IVNNs represented by  $[T_a^L, T_a^U], [I_a^L, I_a^U], [F_a^L, F_a^U]$  and  $[T_b^L, T_b^U], [I_b^L, I_b^U], [F_b^L, F_b^U]$ , respectively. Following mathematical operations are applied between two IVNNs a and b as defined below [68]:

$$a^{c} = \langle [T_{a}^{L}, T_{a}^{U}], [1 - I_{a}^{L}, 1 - I_{a}^{U}], [F_{a}^{L}, F_{a}^{U}] \rangle,$$
(2)

 $a \subseteq b$  if and only if  $a \subseteq b$  and only if

$$T_{a}^{L} \leq T_{b}^{L}, T_{a}^{U} \leq T_{b}^{U}; I_{a}^{L} \geq I_{b}^{L}, I_{a}^{U} \geq I_{b}^{U}; F_{a}^{L} \geq F_{b}^{L}, F_{a}^{U} \geq F_{b}^{U},$$
(3)  
d only if  $a \subseteq b$  and  $b \subseteq a$  (4)

a = b if and only if  $a \subseteq b$  and  $b \subseteq a$ 

$$a \oplus b = \langle \begin{bmatrix} T_{a}^{L} + T_{b}^{L} - T_{a}^{L} T_{b}^{L}, T_{a}^{U} + T_{b}^{U} - T_{a}^{U} T_{b}^{U} \end{bmatrix}, \\ \begin{bmatrix} I_{a}^{L} I_{b}^{L}, I_{a}^{U} I_{b}^{U} \end{bmatrix}, \begin{bmatrix} F_{a}^{L} F_{b}^{L}, F_{a}^{U} F_{b}^{U} \end{bmatrix}, \end{pmatrix}$$
(5)

$$a \otimes b = \langle \begin{bmatrix} T_{a}^{L} T_{b}^{L}, T_{a}^{U} T_{b}^{U} \end{bmatrix}, \begin{bmatrix} I_{a}^{L} + I_{b}^{L} - I_{a}^{L} I_{b}^{L}, I_{a}^{U} + I_{b}^{U} - I_{a}^{U} I_{b}^{U} \end{bmatrix}, \\ \begin{bmatrix} F_{a}^{L} + F_{b}^{L} - F_{a}^{L} F_{b}^{L}, F_{a}^{U} + F_{b}^{U} - F_{a}^{U} F_{b}^{U} \end{bmatrix} \rangle.$$
(6)

**Definition 2:** A set of IVNNs is represented by the notation  $x_k =$  $\langle [T_k^L, T_k^U], [I_k^L, I_k^U], [F_k^L, F_k^U] \rangle$ , where k is the expert (k = 1, 2, 3, ..., K). To aggregate multiple IVNNs, the interval valued neutrosophic weighted arithmetic operator (INNWA) is defined as follows [68]:

$$INNWA(x_{1}, x_{2}, ..., x_{K}) = \sum_{k=1}^{K} y_{k} x_{k} = \left[1 - \prod_{k=1}^{K} (1 - T_{k}^{L})^{y_{k}}, 1 - \prod_{k=1}^{K} (1 - T_{k}^{U})^{y_{k}}\right], \\ \left\langle \left[\prod_{k=1}^{K} (I_{k}^{L})^{y_{k}}, \prod_{k=1}^{K} (I_{k}^{U})^{y_{k}}\right], \left[\prod_{k=1}^{K} (F_{k}^{L})^{y_{k}}, \prod_{k=1}^{K} (F_{k}^{U})^{y_{k}}\right]\right\rangle,$$
(7)

where  $y_k$  denotes the weight vector of the experts.

**Definition 3:** The deneutrosophication function of an IVNN *a* is calculated using Eq. (8) [69]:

$$K(a) = \left(\frac{\left(T_{a}^{L} + T_{a}^{U}\right)}{2} + \left(1 - \frac{\left(I_{a}^{L} + I_{a}^{U}\right)}{2}\right)\left(I_{a}^{U}\right) - \left(\frac{\left(F_{a}^{L} + F_{a}^{U}\right)}{2}\right)\left(1 - F_{a}^{U}\right)\right),\tag{8}$$

where  $a = \langle [T_a^L, T_a^U], [I_a^L, I_a^U], [F_a^L, F_a^U] \rangle$ .

**Definition 4:** Let *a* and *b* be two IVNNs represented by  $\begin{bmatrix} T \\ a \end{bmatrix}$ ,  $\begin{bmatrix} I \\ a \end{bmatrix}$ ,  $\begin{bmatrix} I \\ a \end{bmatrix}$ ,  $\begin{bmatrix} F \\ a$ and  $\begin{bmatrix} T_{b}^{L}, T_{b}^{U} \end{bmatrix}, \begin{bmatrix} I_{b}^{L}, I_{b}^{U} \end{bmatrix}, \begin{bmatrix} F_{b}^{L}, F_{b}^{U} \end{bmatrix}$ , respectively. The Euclidean distance between two IVNNs are given by Eq. (9) [70]:

$$d_{E}(a,b) = \sqrt{\frac{1}{6} \left( \left( \left( T_{a}^{L} - T_{b}^{L} \right)^{2} + \left( T_{a}^{U} - T_{b}^{U} \right)^{2} + \left( I_{a}^{L} - I_{b}^{L} \right)^{2} + \left( I_{a}^{U} - I_{b}^{U} \right)^{2} + \left( F_{a}^{L} - F_{b}^{L} \right)^{2} + \left( F_{a}^{U} - F_{b}^{U} \right)^{2} + \left( F_{a}^{U} - F_{b}^{U} \right)^{2} \right) \right)}.$$
(9)

Definition 5: The function in Eq. (10) returns the Euclidean distance if the difference between two deneutrosophicated IVNNs is greater than zero, otherwise it returns zero.

$$Z(a,b) = \begin{cases} d_E(a,b), & \text{if } K(a) - K(b) > 0, \\ 0, & \text{if } K(a) - K(b) \le 0, \end{cases}$$
(10)

where  $a = [T_{a}^{L}, T_{a}^{U}], [I_{a}^{L}, I_{a}^{U}], [F_{a}^{L}, F_{a}^{U}], b = [T_{b}^{L}, T_{b}^{U}], [I_{b}^{L}, I_{b}^{U}], [F_{b}^{L}, F_{b}^{U}].$ 

#### **3.2 Interval Valued Neutrosophic EDAS Method**

This section presents the steps of the proposed IVN EDAS method. The proposed method addresses a group decision making problem that is under the assumption that there are the set of *n* alternatives  $X = \{X_1, X_2, ..., X_i, ..., X_n\}$  evaluated based on *m* criteria  $C = \{C_1, C_2, ..., C_j, ..., C_m\}$  by *K* DMs  $D = \{D_1, D_2, ..., D_k, ..., D_K\}$ . In the rest of this section, the steps of the proposed IVN EDAS method are presented.

**Step 1:** The linguistic evaluations of each expert are collected regarding the alternatives' performance with respect to the criteria. These linguistic evaluations are converted to IVNNs by using the IVN scale given in Table 1 to obtain the IVN matrix of each expert as presented in Table 2, where  $x_{ijk} = \langle \begin{bmatrix} T \\ ijk \end{bmatrix}, \begin{bmatrix} I \\ ijk \end{bmatrix}, \begin{bmatrix} F \\ ijk \end{bmatrix}, \begin{bmatrix} I \\ ijk \end{bmatrix}, \begin{bmatrix} I$ 

L	inguistic Terms	$\langle T, I, F \rangle$		
CL	Certainly Low	<pre>([0.05, 0.2], [0.6, 0.7], [0.75, 0.9])</pre>		
VL	Very Low	<pre>([0.15, 0.3], [0.5, 0.6], [0.65, 0.8])</pre>		
L	Low	([0.25, 0.4], [0.4, 0.5], [0.55, 0.7])		
BA	Below Average	([0.35, 0.5], [0.3, 0.4], [0.45, 0.6])		
А	Average	([0.40, 0.6], [0.1, 0.2], [0.40, 0.6])		
AA	Above Average	<pre>([0.45, 0.6], [0.3, 0.4], [0.35, 0.5])</pre>		
Н	High	([0.55, 0.7], [0.4, 0.5], [0.25, 0.4])		
VH	Very High	<pre>([0.65, 0.8], [0.5, 0.6], [0.15, 0.3])</pre>		
CH	Certainly High	([0.75, 0.9], [0.6, 0.7], [0.05, 0.2])		

Table 1 IVN Scale for alternative evaluation [45]

**Table 2** IVN decision matrix of an expert k

Crit.	AL <sub>1</sub>	$AL_n$
C <sub>1</sub>	$[T^L_{11k}, T^U_{11k}], [I^L_{11k}, I^U_{11k}], [F^L_{11k}, F^U_{11k}]$	$[T^L_{n1k}, T^U_{n1k}], [I^L_{n1k}, I^U_{n1k}], [F^L_{n1k}, F^U_{n1k}]$
÷	:	:
$C_m$	$[T_{1mk}^L, T_{1mk}^U], [I_{1mk}^L, I_{1mk}^U], [F_{1mk}^L, F_{1mk}^U]$	$[T_{nmk}^L, T_{nmk}^U], [I_{nmk}^L, I_{nmk}^U], [F_{nmk}^L, F_{nmk}^U]$

**Step 2:** To obtain the aggregated IVN decision matrix, the IVN decision matrix of the experts are aggregated by using the INNWA operator defined in Eq. (7). Table 3 shows the representation of the aggregated IVN decision matrix  $x_A$ , where  $x_A = [x_{ijA}]_{n*m} = \langle [T_{ijA}^{L}, T_{ijA}^{U}], [I_{ijA}^{L}, I_{ijA}^{U}] [F_{ijA}^{L}, F_{ijA}^{U}] \rangle$  represents the aggregated evaluation of the experts for the alternative i with respect to criterion *j*.

**Step 3:** Each expert k evaluates the criteria importance with linguistic terms. These evaluations are converted to IVNNs using the scale given in Table 4. The following criteria weights matrix is obtained for each expert k:

$$W = \left[ w_{jk} \right]_{1*m} \tag{11}$$

 Table 3 Aggregated IVN decision matrix

Crit.	$AL_1$	$AL_n$
<i>C</i> <sub>1</sub>	$[T^L_{11A}, T^U_{11A}], [I^L_{11A}, I^U_{11A}], [F^L_{11A}, F^U_{11A}]$	$[T_{n1A}^L, T_{n1A}^U], [I_{n1A}^L, I_{n1A}^U], [F_{n1A}^L, F_{n1A}^U]$
:	÷	÷
$C_m$	$[T_{1mA}^L, T_{1mA}^U], [I_{1mA}^L, I_{1mA}^U], [F_{1mA}^L, F_{1m}^U]$	$[T_{nmA}^{L}, T_{nmA}^{U}], [I_{nmA}^{L}, I_{nmA}^{U}], [F_{nmA}^{L}, F_{nmA}^{U}]$

**Step 4:** The criteria weights are aggregated by INNWA in Eq. (7) and normalized to obtain the aggregated criteria weights matrix  $W_A$  given in Eq. (12), where  $w_j$  represents the aggregated weight of the criterion j.

$$W_A = \left[ w_j \right]_{1 * m} \tag{12}$$

**Step 5:** The average solution matrix AV is obtained by taking the average of the aggregated alternative evaluations  $x_{iiA}$  as given in Eq. (13):

$$AV = [AV_j]_{1*m} = \frac{1}{n} \sum_{i=1}^{n} x_{ijA}$$
(13)

Table 4 IVN Scale for criteria evaluation [45]

	Linguistic Terms	$\langle T, I, F \rangle$
CLI	Certainly Low Importance	<pre>([0.05, 0.25], [0.6, 0.7], [0.75, 0.95])</pre>
VLI	Very Low Importance	<pre>([0.15, 0.35], [0.5, 0.6], [0.65, 0.85])</pre>
LI	Low Importance	<pre>([0.25, 0.45], [0.4, 0.5], [0.55, 0.75])</pre>
BAI	Below Average Importance	<pre>([0.35, 0.55], [0.3, 0.4], [0.45, 0.65])</pre>
AI	Average Importance	([0.40, 0.60], [0.1, 0.2], [0.40, 0.60])
AA	Above Average Importance	<pre>([0.45, 0.65], [0.3, 0.4], [0.35, 0.55])</pre>
HI	High Importance	([0.55, 0.75], [0.4, 0.5], [0.25, 0.45])
VHI	Very High Importance	([0.65, 0.85], [0.5, 0.6], [0.15, 0.35])
CHI	Certainly High Importance	<pre>([0.75, 0.95], [0.6, 0.7], [0.05, 0.25])</pre>

**Step 6:** Each alternative is classified as either positively positioned or negatively positioned to the average solution  $AV_j$  with respect to each criterion. Positively positioned alternatives signify a superior performance than the average solution in terms of a criterion and contrarily, the negatively positioned alternatives indicate a worse than the average performance. The positive distance to average (PDA) of a positively positioned alternative must be positive and negative distance to average (NDA) must be zero. Contrarily, for the negatively positioned alternatives PDA must be zero and NDA must be positive. The calculation of the positive and negative distance to the average matrices (PDA and NDA) are modified by eliminating the neutrosophic subtraction to preserve the characteristics of IVNNs and to avoid negative numbers in the IVN parameters. The PDA and NDA are

calculated as defined in Eqs. (14) and (15) based on the criterion type, respectively. The Euclidian distance is calculated if the difference between the deneutrosophicated aggregated alternative performance  $K(x_{ijA})$  and the deneutrosophicated average solution  $AV_i$  is positive, otherwise it assigns zero.

$$PDA = \left[pda_{ij}\right]_{m*n} = \begin{cases} z\left(K(x_{ijA}) - K(AV_j)\right) & \text{if } j \in B, \\ z\left(K(AV_j) - K(x_{ijA})\right) & \text{if } j \in C, \end{cases}$$
(14)

$$NDA = \left[nda_{ij}\right]_{m*n} = \begin{cases} z\left(K(AV_j) - K(x_{ijA})\right) & \text{if } j \in B, \\ z\left(K(x_{ijA}) - K(AV_j)\right) & \text{if } j \in C, \end{cases}$$
(15)

where B and C denote the set of the benefit and cost criteria, respectively and z(.) is a function defined in Eq. (10).

**Step 7:** The weighted total positive distance and negative distance from the average  $sp_i$  and  $sn_i$  are calculated for each alternative *i* by using Eqs. (16) and (17):

$$sp_i = \sum_{j=1}^m (w_j * pda_{ij}), \tag{16}$$

$$sn_i = \sum_{j=1}^{m} (w_j * nda_{ij}).$$
 (17)

**Step 8:** The weighted total positive distance and negative distance from the average values are normalized as given in Eqs. (18) and (19):

$$nsp_i = \frac{sp_i}{Max(sp_i)},\tag{18}$$

$$nsn_i = 1 - \frac{sn_i}{Max(sn_i)}.$$
 (19)

**Step 9:** To obtain the appraisal score  $as_i$ , the average of the normalized positive distance and normalized negative distance is computed as follows:

$$as_i = \frac{1}{2}(nsp_i + nsn_i).$$
<sup>(20)</sup>

Step 10: The alternatives are ranked in descending order of the appraisal score.

### 4. CASE STUDY

The methodology of this study consists of three phases, namely the preparation process, the application of the IVN EDAS method and the validation of results. Figure 1 illustrates the flowchart of the presented methodology. This section presents the steps of the methodology implemented for the case study.

### 4.1. Problem Definition

The paper aims to determine the most appropriate drone for monitoring and detection of forest fires. The drone alternatives should be evaluated with respect to multiple and conflicting criteria. Although the drone characteristics and performances based on the

criteria are known as its own specifications, the actual performance of a drone under mission highly depends on the environmental circumstances. This aspect brings the need for the use of fuzzy logic. The criteria were established through research of articles and books, as well as interviews with experts in the field. The authors and the experts omitted the sub-criteria that have similar or the exact opposite meaning in order to avoid overlapping in the criteria. For instance, the charging time and the battery life have rather opposing denotations as one refers to the time during which a drone cannot operate while the other signifies its up-time. As a result, 14 sub-criteria have been identified in total and presented in Table 6.

After determining the criteria and sub-criteria, a survey was conducted among expert DMs in the field to weigh the criteria and alternatives. All DMs were given different weights according to their expertise in the process of selecting a drone for detecting forest fires. The importance weights of the experts are assigned based on their experience by the authors. Regarding the type of drones that have been selected for the evaluation in this study, the rotorcraft drones are examined. It has been found that rotorcraft (multirotor) drones tend to be better off in terms of maneuverability, hovering capability, and looser requirements on takeoff and landing than fixed-wing drones [71]. Therefore, four different drone models of multirotor type selected as potential alternatives for detecting forest fires, and their technical specifications are presented in Table 7. The information on the experts and their weights are given as follows:



Fig. 1 Flowchart of the methodology

**DM1 (40%):** The first expert is a retired F16 pilot who provides drone training. He has been providing such training for about six years. He has a comprehensive knowledge on the aviation regulations, safety standards, and 10+ years of flight experience with drones.

**DM2** (25%): The second expert is a fire chief. Although not fully experienced with drones, he had first-hand experience with forest fires for over 15 years. Thus, he has an indepth knowledge of emergency response, safety protocols and fire behavior, especially how fires spread and the factors impacting the fire behavior. He mainly contributed by assessing the importance of drone attributes.

**DM3** (**35%**): The third expert is a commercial drone user with an experience over 10 years. He has an in-depth knowledge of the technical features such as drone models, sensors, cameras, payloads, its capabilities, and the areas of drone usage. Further, he has an extensive knowledge on the sensor operation.

The description of each criterion is given as follows:

Max. Takeoff Weight (C1.1) (Benefit): It represents the maximum weight of a drone at which it is allowed to take off. In addition to its main body, for the needs of specific usage drones can be equipped with sensors, cameras, communication tools, and components such as fire extinction equipment.

Max. Horizontal Speed (C1.2) (Benefit): It expresses the maximum speed at which a drone can fly in the horizontal plane.

**Max. Hover Time (C1.3) (Benefit):** It is the maximum time during which a drone hovers and holds its position in the air.

Max. Flight Time (C1.4) (Benefit): It is the criterion that shows the total flight time of a drone during a single takeoff and landing.

**Charging Time (C1.5) (Benefit):** It indicates how long it takes a fully discharged battery of a drone to charge completely.

Max. Operating Altitude (C1.6) (Benefit): It is the maximum altitude at which a drone can operate above sea level.

Criteria	Sub-Criteria	References
	C1.1. Max. Takeoff Weight	[71, 72, 73, 74, 75]
	C1.2. Max. Horizontal Speed	[28, 59, 60, 73, 74, 76]
	C1.3. Max. Hover Time	[71]
	C1.4. Max. Flight Time	[59, 71, 73, 77]
C1. Technical	C1.5. Charging Time	[28, 77]
Abilities	C1.6. Max. Operating Altitude	[59, 60, 72, 73, 76, 78]
	C1.7. Wind Resistance	[79, 80]
	C1.8. Ingress Protection Rating of Drones	[77, 79]
	C1.9. Operation Frequency	[74, 81, 82]
	C1.10. Hovering Accuracy	[77]
C2 Vision	C2.1. Zoom Camera	[72, 77, 78, 83]
C2. VISIOII-	C2.2. Resolution	[77, 78, 84]
Tashnologias	C2.3.Thermal Camera Accuracy	[19, 77, 83, 85]
rechnologies	C2.4. Obstacle Sensors (Range)	[28, 77, 86]

 Table 6 Criteria List for the drone selection for forest fires

Wind Resistance (C1.7) (Benefit): The wind resistance of a drone can be affected by its weight and size. The wind resistance level is given as a value between 0 and 12. The greater the wind resistance level, the more stable can a drone operate in high-wind circumstances [80].

**Ingress Protection Rating of Drones (C1.8) (Benefit):** This criterion gives the IP value of a drone. IP Code (also known as International Protection Rating) categorizes the degrees of protection given in electrical enclosures against solid object infiltration, dust, accidental touch, and liquids. The IP Code consists of two digits. The first digit ranges from 1 to 6 and signifies the protection against solids, whereas the second digit ranges from 1 to 8 and indicates the resistance against liquids.

**Operation Frequency (C1.9) (Benefit):** It represents the frequency band that a drone uses for communication and data download.

Drone Alternative	Matr	rice 30	Matrice 30	Matri R'	се 210 ГК	Phantom 4 RTK		
Max. Takeoff Weight (g) (C1.1)	2	38	930	1230		-		
Max. Hor. Speed (m/s) (C1.2)	2	23	23		2	20	1	6
Max. Hover Time (min) (C1.3)	3	36	50		2	24	2	:6
Max. Flight Time (min) (C1.4)	2	41	55		3	3	3	0
Charging Time (min) (C1.5)	30-	- 50	70		7	0	3	0
Max. Operating Altitude (m) (C1.6)	70	000	5000		3000		6000	
Wind Resistance (m/s) (C1.7)	1	5	15	12		10		
Ingress Protection (C1.8)	IF	255	IP45		IP43		IP	42
Operation Frequency (C1.9)	2.4 G	Hz or 5 Hz	2.4 GHz or 5 GHz		2.4 - 2.48 GHz; 5.7 - 5.8 GHz		2.4 GHz or 5 GHz	
Hovering Accuracy (C1.10)	V:0.1m	H: 0.3m	V: 0.1m	H: 0.1m	V: 0.1m	H: 0.1m	V: 0.1m	H: 0.5m
Zoom Camera (C2.1)	5	m	2 m	1	1 m		1 m	
Resolution (C2.2)	3840	x2160	1920x1	080	1920:	x1080	3840	x2160
Thermal Camera Accuracy (C2.3)	±2°Co	or ±2%	±5°Cor ±5%		±7°Co	or ±7%		-
Obstacle Sensors (C2.4)	F:0.6- 38m	U, D, B, S: 0.5- 33m	F/B/L/R:0.7- 40m	U/D: 0.6- 30m	F: 0.7-40 m	U, D, B, S: 0.7-40 m	0.7-3	30 m

 Table 7 Technical specifications of drone alternatives

**Hovering Accuracy (C1.10) (Benefit):** It indicates how stable a drone stays while hovering. This criterion is given in meters (m) by how much a drone deviates from its initial position in vertical and horizontal directions.

Zoom Camera (C2.1) (Benefit): This criterion represents how far the embedded visual camera of a drone can take a clear image.

**Resolution (C2.2) (Benefit):** Each visual camera has a resolution value. The resolution is one of the factors that directly affect image quality.

**Thermal Camera Accuracy (C2.3) (Benefit):** Thermal cameras do not need any light source for imaging. In the context of forest fires, thermal camera imaging is especially important for early detection.

**Obstacle Sensors (Range) (C2.4) (Benefit):** It indicates the furthest distance from which a drone can detect the obstacles.

Although the cost of a drone has been generally included as a criterion in the studies dealing with the drone selection problem for both military and civilian purposes, in this study, it has been excluded due to two reasons – first, the economic aspect must be of minor importance in disaster management as it is critical to human life and a cause of great economic and environmental losses. Second, governments do not make public procurement decisions solely based on the economic perspective, instead sustainable and strategic options are preferred.

As mentioned previously, there are interrelationships between the technical features of a drone as well as some drone attributes are highly affected by certain environmental circumstances. To give a few examples of the tradeoffs between technical internal attributes of a drone, the relationship between energy consumption and flight speed is well known. In a lower speed range, the speed does not affect the power consumption significantly. Yet in a relatively higher speed range, the power consumption increases exponentially [29]. Increased power consumption during operation means a drone requires to get charged more frequently and higher downtime which lowers the overall system efficiency. Likewise, the payload of a drone has a significant effect on power consumption. Increasing the payload of a drone rises the power consumption significantly and lowers the maximum flight time of a drone [43]. Besides the interrelation among internal features of drones, their performance is also highly affected by environmental circumstances such as temperature, wind, and precipitation. Drones with relatively small sizes and weights are prone to wind disturbances [87]. Further, it has been found that high temperature has little effect on the flight performance of a drone, but it significantly shortens the lifetime and even damages the battery, whereas extremely low temperature worsens the flight and the battery performance [88]. It can be concluded that there is a highly complex structure (1) between the internal attributes of a drone, (2) between the environmental conditions and the internal attributes of a drone. As a result, it is hard to describe and quantify the model with clear and exact terms and relations.

We conclude that the discussed problem involves uncertainties such as doubtfulness or vagueness, ambiguity, and inconsistency among its elements. The experts submit their subjective opinions based own personal experiences, background, and other individual characteristics. Further, the conflicting and complex relationships between the criteria, which were explained above, contribute to the ambiguous nature of the problem. To represent the vagueness in the decision-making process and the conflicting criteria in the problem, the IVN EDAS method is preferred.

### 4.2. Numerical Application

The application of the method is illustrated by the evaluation of four potential drone models, that are identified as suitable for forest surveillance and fire detection. The drone models are assessed based on the pre-determined 14 criteria by the three DMs. The steps of the methodology are given with the obtained results as follows:

**Step 1:** In the first step, the linguistic evaluations of the experts are collected (Table 8) and the IVN decision matrices are constructed by using the IVN scale in Table 1.

**Step 2:** The IVN evaluations of the alternatives are aggregated using Eq. (7). Table 9 presents the aggregated IVN matrix for Alternative 1.

**Steps 3-4:** The linguistic evaluations for criteria weights are collected, which are converted to IVNNs by using Table 4. The IVN matrices of the experts are then aggregated by using INNWA operator. The linguistic evaluations and the average IVN weights of the criteria are presented in Table 10.

**Step 5:** The average solution matrix is calculated by Eq. (13). Table 11 shows the average solution matrix.

DM1					DI	M2			DI	М3		
Criteria	AL1	AL2	AL3	AL4	AL1	AL2	AL3	AL4	AL1	AL2	AL3	AL4
C1.1	Н	А	AA	L	Η	BA	AA	L	L	CH	AA	BA
C1.2	Η	Η	AA	CH	Н	Η	AA	CH	CH	Η	AA	CH
C1.3	CH	CH	Н	AA	CH	Н	AA	Α	Η	CH	Α	А
C1.4	CH	Н	AA	Α	CH	Н	AA	Α	Η	CH	Α	А
C1.5	Η	AA	Α	Α	Η	AA	Α	Α	Η	L	BA	CH
C1.6	AA	CH	А	Н	AA	CH	BA	Н	Н	А	L	AA
C1.7	AA	AA	AA	Α	AA	AA	Α	BA	А	Α	BA	L
C1.8	AA	CH	Α	BA	AA	CH	Α	BA	Н	Α	Α	BA
C1.9	AA	Η	AA	AA	AA	AA	AA	AA	Н	Η	Η	Η
C1.10	AA	Н	AA	AA	AA	Н	Α	Α	AA	Α	Α	А
C2.1	CH	Η	AA	Α	Н	Η	AA	Α	AA	Η	Α	Α
C2.2	Η	CH	Н	AA	Η	CH	Н	AA	CH	Α	BA	BA
C2.3	Η	CH	А	BA	AA	CH	Α	BA	Η	AA	AA	BA
C2.4	Η	AA	BA	Α	Η	AA	Η	Α	Η	Η	Η	Α

Table 8 Linguistic evaluations of the experts

**Step 6:** The PDA and NDA matrices are obtained and presented in Table 12. The calculation of PDA and NDA values for Alternative 1 with respect to Criterion 3, which is a benefit criterion, is as follows:

$$pda_{13} = z(K(x_{13A}) - K(AV_3)), \ nda_{13} = z(K(AV_3) - K(x_{13A})),$$
  

$$K(x_{13A}) = 0.9120; \ K(AV_3) = 0.7474; \ pda_{13} = z(0.1646) = d_E,$$
  

$$(x_{13A}, AV_3) = 0.1344; \ nda_{13} = z(K(AV_3) - K(x_{13A})) = 0.$$

 Table 9 Aggregated IVN decision matrix of Alternative 1

C1.1	[(0.462,0.618), (0.400,0.500), (0.329,0.487)]
C1.2	[(0.634, 0.796), (0.461, 0.562), (0.142, 0.314)]
C1.3	[(0.693, 0.853), (0.521, 0.622), (0.088, 0.255)]
C1.4	[(0.693, 0.853), (0.521, 0.622), (0.088, 0.255)]
C1.5	[(0.550, 0.700), (0.400, 0.500), (0.250, 0.400)]
C1.6	[(0.487, 0.638), (0.332, 0.432), (0.311, 0.462)]
C1.7	[(0.433, 0.600), (0.204, 0.314), (0.367, 0.533)]
C1.8	[(0.487, 0.638), (0.332, 0.432), (0.311, 0.462)]
C1.9	[(0.487, 0.638), (0.332, 0.432), (0.311, 0.462)]
C1.10	[(0.450, 0.600), (0.300, 0.400), (0.350, 0.500)]
C2.1	[(0.618, 0.786), (0.425, 0.529), (0.148, 0.328)]
C2.2	[(0.634, 0.796), (0.461, 0.562), (0.142, 0.314)]
C2.3	[(0.527, 0.678), (0.372, 0.473), (0.272, 0.423)]
C2.4	[(0.550, 0.700), (0.400, 0.500), (0.250, 0.400)]

 $[(T^{L}, T^{U}), (I^{L}, I^{U}), (F^{L}, F^{U})]$ 

Table 10 Linguistic criteria evaluations and aggregated IVN criteria weights

				Aggregated criteria weights
Criteria	DM1	DM2	DM3	$[(T^{L}, T^{U}), (I^{L}, I^{U}), (F^{L}, F^{U})]$
C1.1 Max. Takeoff Weight	VL	BA	BA	[(0.276,0.479), (0.368,0.470), (0.521,0.724)]
C1.2 Max. Horizontal Speed	Н	А	А	[(0.465,0.669), (0.174,0.289), (0.331,0.535)]
C1.3 Max. Hover Time	L	BA	А	[(0.331,0.532), (0.229,0.343), (0.468,0.669)]
C1.4 Max. Flight Time	Н	Н	VH	[(0.588,0.791), (0.432,0.533), (0.209,0.412)]
C1.5 Charging Time	VL	VL	L	[(0.186,0.387), (0.462,0.563), (0.613,0.814)]
C1.6 Max. Operating Altitude	L	VH	А	[(0.427,0.644), (0.260,0.380), (0.356,0.573)]
C1.7 Wind Resistance	Н	AA	AA	[(0.492,0.694), (0.337,0.437), (0.306,0.508)]
C1.8 Ingress Protection Rating of Drones	VH	VH	Н	[(0.618,0.821), (0.462,0.563), (0.179,0.382)]
C1.9 Operation Frequency	А	BA	L	[(0.338,0.539), (0.214,0.328), (0.461,0.662)]
C1.10 Hovering Accuracy	VL	VL	L	[(0.186,0.387), (0.462,0.563), (0.613,0.814)]
C2.1 Zoom Camera	А	AA	Η	[(0.469,0.672), (0.214,0.328), (0.328,0.531)]
C2.2 Resolution	AA	А	Н	[(0.476,0.678), (0.252,0.364), (0.322,0.524)]
C2.3 Th. Camera Accuracy	VH	VH	VH	[(0.650, 0.850), (0.500, 0.600), (0.150, 0.350)]
C2.4 Obstacle Sensors (Range)	BA	А	VL	[(0.300,0.503), (0.273,0.388), (0.497,0.700)]

Drone Selection for Forest Surveillance and Fire Detection using Interval Valued Neutrosophic EDAS... 19

	$[(T^L, T^U), (I^L, I^U), (F^L, F^U)]$
C1.1	[(0.437,0.599), (0.327,0.433), (0.348,0.515)]
C1.2	[(0.596, 0.749), (0.440, 0.541), (0.198, 0.353)]
C1.3	[(0.575, 0.741), (0.362, 0.468), (0.216, 0.384)]
C1.4	[(0.540, 0.712), (0.321, 0.425), (0.249, 0.425)]
C1.5	[(0.470, 0.640), (0.266, 0.374), (0.318, 0.493)]
C1.6	[(0.501, 0.664), (0.307, 0.419), (0.289, 0.455)]
C1.7	[(0.402, 0.570), (0.213, 0.323), (0.397, 0.564)]
C1.8	[(0.474, 0.644), (0.263, 0.371), (0.316, 0.489)]
C1.9	[(0.497, 0.648), (0.342, 0.443), (0.301, 0.453)]
C1.10	[(0.448, 0.617), (0.214, 0.323), (0.351, 0.519)]
C2.1	[(0.500, 0.672), (0.282, 0.386), (0.291, 0.465)]
C2.2	[(0.550, 0.711), (0.361, 0.469), (0.234, 0.400)]
C2.3	[(0.491, 0.654), (0.322, 0.426), (0.301, 0.465)]
C2.4	[(0.479, 0.643), (0.297, 0.397), (0.319, 0.483)]

 Table 11 The average solution matrix

Table 12 The positive distance to average and negative distance to average matrices

		PDA		1	NDA			
Criterion	AL1	AL2	AL3	AL4	AL1	AL2	AL3	AL4
C1.1	0.0446	0.1129	0	0	0	0	0.0191	0.1291
C1.2	0.0392	0	0	0.1544	0	0.0459	0.1458	0
C1.3	0.1344	0.1523	0	0	0	0	0.1111	0.1756
C1.4	0.1718	0.1141	0	0	0	0	0.1121	0.1755
C1.5	0.0973	0	0	0.0947	0	0.0797	0.1022	0
C1.6	0	0.1399	0	0.0311	0.0191	0	0.1457	0
C1.7	0.0256	0.0256	0.0089	0	0	0	0	0.0519
C1.8	0.0397	0.1659	0	0	0	0	0.1174	0.1070
C1.9	0	0.0298	0	0	0.0099	0	0.0099	0.0099
C1.10	0.0483	0.0495	0	0	0	0	0.0414	0.0414
C2.1	0.1339	0.0776	0	0	0	0	0.0722	0.1372
C2.2	0.0903	0.0989	0	0	0	0	0.0543	0.1198
C2.3	0.0390	0.1765	0	0	0	0	0.1186	0.1192
C2.4	0.0828	0.0224	0.0351	0	0	0	0	0.1329

**Steps 7-10:** The PDA and NDA values are weighted by the criteria weights and summed over the criteria to obtain the positive and negative weighted total distance to the average  $sp_i$  and  $sn_i$ , respectively. Then they are normalized to calculate  $nsp_i$  and  $nsn_i$ . The appraisal score of each alternative is calculated and the alternatives are ranked in descending order of the appraisal score.

				Rank
	$nsp_i$	nsn <sub>i</sub>	$as_i$	
AL1	0.7458	0.9785	0.8621	2
AL2	1	0.9191	0.9595	1
AL3	0.0299	0.1176	0.0738	4
AL4	0.1967	0	0.0983	3

Table 13 Weighted total distances, appraisal scores and final ranking of the alternatives

The rankings presented in Table 13 demonstrate that Matrice 300 RTK (AL2) has been evaluated as the most preferable drone with an appraisal score of 0.9595 and is closely followed by Matrice 30 (AL1) with an AS of 0.8621. The minor difference of ranking scores between AL1 and AL2 should be interpreted as both drone models provide relatively comparable performances. The final ranking obtained is as follows: AL2 > AL1 > AL4 > AL3.

### 4.3. Sensitivity Analysis

An extensive one-at-a-time sensitivity analysis is conducted to check for the robustness of the results to the change in the criteria weights. In total 84 sets of criteria weights are built following the approach in Gorcun et al. [89]. In each scenario, the initial weight of a criterion is decreased by the predetermined modification degree v (15%, 30%, 45%, 60%, 75%, 90%), and the weights of the rest of the criteria are increased equally while satisfying that the sum of the weights equals to 1. Figure 2 illustrates the criteria weights used in the scenarios from SC-57 to SC-70 which are formed by a 75%-modification degree, and Figure 3 shows the resulting rankings of the alternatives in the generated 84 scenarios.



Fig. 2 The sets of criteria weights at 75% modification of each criterion



Fig. 3 Rankings of the alternatives generated through 84 scenarios

The result of the sensitivity analysis demonstrates that approximately a minimum modification by 30% of Criterion 1.2, which corresponds to scenarios SC-16, SC-30, SC-44, SC-58 and SC-72, results in the third and fourth alternatives to change ranking positions. The same result emerges under the minimum approximate modification by 45% of Criterion 1.5 and Criterion 1.6. Concludingly, in 100% of the scenarios, Alternative 2 and Alternative 1 preserved the first and second ranking positions, respectively, and in 84.5% of the scenarios (71 out of 84 scenarios) Alternative 4 has been ranked at the third and Alternative 3 at the fourth position. The sensitivity analysis indicates that the results regarding the first two ranking positions are robust against the weight changes in all criteria. The latter two positions are slightly sensitive to the changes in C1.2, C1.5 and C1.6.

### 4.4. Comparative Analysis

In order to check for the veracity of the method, a comparative analysis is conducted by using two MCDM methods, the interval valued neutrosophic TOPSIS, and interval valued neutrosophic CODAS methods. We preferred to apply the distance-based methods for the comparison in accordance with the method of application in this paper. For the application of IVN TOPSIS, we adopted the steps given in Karasan et al. [90]. The criteria weights are taken as given in Table 10. The Euclidian and Hamming distance measures used in this section are calculated by adopting the steps presented in the study of [91] and both equations are modified by the summation over the criteria and the division by the number of criteria in the model.

In the applied IVN TOPSIS method, first, the decision matrices are aggregated by weighting with the priorities of DMs to obtain the aggregated neutrosophic decision matrix. Then the weighted normalized decision matrix is computed by weighting the decision matrix with criteria weights. The positive ideal and the negative ideal solution are calculated as presented in Table 14. Based on the determined ideal solutions, the Euclidian distance of each alternative to the PIS and NIS is computed.

As for the second method selected for the comparative analysis, the steps of IVN CODAS are adopted from the study of [95] with minor changes from their method. The threshold parameter ( $\theta$ ) used in the threshold function is taken as 0.005 since the differences between the Euclidian distances of alternatives to the negative ideal solution are significantly close. To incorporate the difference of Hamming distances into the calculation, we adjusted the value of the threshold parameter. Table 15 shows the results obtained by IVN TOPSIS, IVN CODAS and IVN EDAS methods.

The IVN EDAS method gives the same ranking with the IVN TOPSIS and IVN CODAS methods indicating that the proposed IVN EDAS method provides veracious and consistent results with other distance based MCDM methods.

	PIS	NIS			
C1.1	[(0.027, 0.037), (0.012, 0.018), (0.01,	[(0.014, 0.022), (0.02, 0.025), (0.025,			
	0.02)]	0.033)]			
C1.2	[(0.074, 0.089), (0.03, 0.039),	[(0.044, 0.059), (0.059, 0.069), (0.035,			
	(0.005,0.02)]	0.049)]			
C1.3	[(0.05, 0.061), (0.011, 0.018), (0.005, 0.011)]	[(0.029, 0.042), (0.038, 0.045), (0.026, 0.045)]			
		0.039)]			
C1.4	[(0.056, 0.069), (0.008, 0.016), (0.007, 0.021)]	[(0.032, 0.048), (0.042, 0.05), (0.032, 0.040)]			
	0.021)]				
C1.5	[(0.018, 0.024), (0.005, 0.008), (0.006, 0.012)]	[(0.012, 0.017), (0.013, 0.016), (0.013, 0.010)]			
	0.013)]	0.019)]			
C1.6	[(0.054, 0.069), (0.018, 0.027), (0.009, 0.024)]	[(0.028, 0.042), (0.03, 0.038), (0.038, 0.052)]			
	(0.024)]	(0.052)			
C1.7	[(0.030, 0.049), (0.017, 0.020), (0.03, 0.044)]	[(0.028, 0.042), (0.019, 0.028), (0.038, 0.052)]			
	(0.044)	[(0,028,0,04),(0,026,0,026),(0,026,0,026)]			
C1.8	[(0.052, 0.000), (0.008, 0.010), (0.008, 0.022)]	[(0.028, 0.04), (0.020, 0.050), (0.050, 0.050)]			
C1.9	[(0.023)]	[(0,035,0,046),(0,027,0,034),(0,023)]			
	[(0.038, 0.049), (0.024, 0.031), (0.02, 0.031)]	[(0.055, 0.040), (0.027, 0.054), (0.025, 0.032)]			
C1.10 C2.1	[(0.016, 0.021), (0.005, 0.008), (0.009)]	[(0,013,0,019),(0,009,0,013),(0,012)]			
	0.015)]	0.018)]			
	[(0.058, 0.074), (0.009, 0.019), (0.014)]	[(0.038, 0.056), (0.04, 0.05), (0.038)]			
	0.031)]	[(0.050), (0.050), (0.050), (0.050),			
C2.2	[(0.06, 0.076), (0.027, 0.036), (0.009,	[(0.038, 0.051), (0.042, 0.051), (0.035]			
	0.027)]	0.048)]			
C2.3	[(0.051, 0.064), (0.011, 0.019), (0.008,	[(0.027.0.038), (0.036, 0.044), (0.034,			
	0.021)]	0.046)]			
C2.4	[(0.033, 0.043), (0.006, 0.012),	[(0.024, 0.036), (0.024, 0.03), (0.024, 0.03)]			
	(0.015.0.024)]	0.036)]			

Table 14 IVN positive ideal solution and negative ideal solution of criteria.

### 4.5 Managerial Implications

The criteria significant for the forest surveillance and fire detection have been established for the first time. In this respect, a framework has been proposed that can be utilized by the authorities and experts that are responsible from the safety and security of

forests against wildfires. On the long run, the integration of drones to the forest surveillance systems can significantly reduce the costs, is more reliable and effective than the other techniques such as satellite based and static WSNs. Therefore, the adoption of the drones as the state-of-the-art technology should be considered and requires a systematic examination with real-world applications.

 Table 15 Comparison of results obtained by IVN TOPSIS, IVN CODAS and IVN EDAS method

	IVN TOPSIS				IVN CODAS		IVN EDAS	
	$D_i^{PIS}$	$D_i^{NIS}$	Closeness coefficient	Rank	Relative assessment	Rank	Appraisal score	Rank
					score			
AL1	0.0103	0.0105	0.0105	2	0.0936	2	0.8621	2
AL2	0.0084	0.0124	0.0124	1	0.1107	1	0.9595	1
AL3	0.0122	0.0088	0.0088	4	-0.1185	4	0.0738	4
AL4	0.0128	0.0093	0.0093	3	-0.0858	3	0.0983	3

### 5. CONCLUSION

In order to deal with forest fires, which is a major cause of deforestation, early warning systems play a crucial role to contain wildfires under control timely. Drones offer superior maneuverability and real-time surveillance capabilities in areas hard to access in comparison to other technologies. Due to the aforementioned benefits, the application of drones is utilized in the early detection systems for forest fires and the type of drones has a significant impact on the effectiveness of these systems. It is noteworthy that the performance of the detection system depends on various factors such as topological characteristics of the terrain, environmental conditions and features of drones [7]. These numerous factors have interrelationships, and some may be conflicting with one another, which can be taken into account through EDAS method.

This study is aimed at addressing the drone selection problem for forest surveillance and fire detection by using the IVN EDAS method thanks to its convenience at handling the conflicting criteria and enriched information representation by the IVN sets. This study contributes by pointing out to the gap that the drone selection problem for forest surveillance and fire detection has been sparsely addressed, by presenting an extensive literature review on the technologies deployed in forest monitoring and fire detection, and by extracting the relevant criteria through an extensive literature review and interviews with the experts in field. As a result, four drone alternatives and 14 criteria have been identified. The alternatives are then evaluated based on the proposed framework using IVN EDAS method. Moreover, a sensitivity analysis is conducted in order to check for robustness by varying the criteria weights. The veracity of the results is validated by carrying out a comparative analysis with IVN TOPSIS method and IVN CODAS method.

The results are in compliance with the study of Pamučar et al. [40] regarding the relative importance of the criteria. The criteria related to the diagnosis and monitoring are assessed as the most critical features in both studies. Further, the ability of stabilization during the flight is evaluated as the second most important in this study. The flight time and the

resolution for monitoring and detection are evaluated as the third and fourth most significant criteria.

It is important to note that the results should be interpreted with caution. The MCDM methods are associated with a major limitation called the rank reversal phenomenon [65]. However, the methodology proposed in this study has not been examined for this limitation and should be checked for in future work.

Based on the findings, the set of criteria and the results should be verified with in-thefield applications and compared with the state of knowledge. In future studies, the performance of fuzzy and neutrosophic MCDM methods should be evaluated by comparing them with the software and simulation results. The criteria set might be extended by consulting more experts and representatives from the authorities. Also, the proposed framework could be implemented under other extensions of fuzzy sets for the comparison of the results.

#### REFERENCES

- Tyukavina, A., Potapov, P., Hansen, M.C., Pickens, A., Stehman, S., Turubanova, S., Parker, D., Zalles, V., Lima, A., Kommareddy, I., Song, X.P., Wang, L., Harris, N., 2022, *Global trends of forest loss due to fire from 2001 to* 2019, Frontiers in Remote Sensing, 3, 825190, doi: 10.3389/frsen.2022.825190
- Institute of Foresters of Australia and Australian Forest Growers, 2020, Submission to the Royal Commission into National Natural Disaster Arrangements. Australian Forestry, 83(3), pp. 107-135.
- Filkov, A., Ngo, T., Matthews, S., Telfer, S., Penman, T.D, 2020, Impact of Australia's catastrophic 2019/20 bushfire season on communities and environment, Retrospective analysis and current trends, Journal of Safety Science and Resilience, 1(1), pp. 44-56.
- Alkhatib, A.A., 2014, A review on forest fire detection techniques. International Journal of Distributed Sensor Networks, 10(3), 597368.
- Chowdary, V., Deogharia, D., Sowrabh, S., Dubey, S., 2022, Forest fire detection system using barrier coverage in wireless sensor networks, Materials Today: Proceedings, 64, pp. 1322-1327.
- Akhloufi, M.A., Couturier, A., Castro, N.A., 2021, Unmanned Aerial Vehicles for wildland fires: Sensing, Perception, Cooperation and Assistance, Drones, 5(1), 15.
- Hristozov, S., Zlateva, P., 2018, Concept model for drone selection in specific disaster conditions, International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences, 42, pp. 245-249.
- 8. Zadeh, L.A., 1965, *Fuzzy sets*, Information and Control, 8(3), pp. 338-353.
- Turksen, I.B., 1986, Interval valued fuzzy sets based on normal forms, Fuzzy Sets and Systems, 20(2), pp. 191-210.
- Smarandache, F., 2006, Neutrosophic set A generalization of the intuitionistic fuzzy set, 2006 IEEE International Conference on Granular Computing, pp. 38-42.
- 11. Atanassov, K., 1986, Intuitionistic fuzzy sets, Fuzzy Sets and Systems, 20(1), pp. 87-96.
- Keshavarz Ghorabaee, M., Zavadskas, E.K., Olfat, L., Turskis, Z., 2015, Multi-criteria inventory classification using a new method of evaluation based on distance from average solution (EDAS), Informatica, 26(3), pp. 435-451.
- Demircan, M.L., Ozcan, B., 2021, A proposed method to evaluate warehouse location for 3PL Cold Chain suppliers in Gulf countries using neutrosophic Fuzzy EDAS, International Journal of Computational Intelligence Systems, 14(1), 202.
- Den Breejen, E., Breuers, M., Cremer, F., Kemp, R., Roos, M., Schutte, K., De Vries, J.S., 1998, Autonomous forest fire detection, Coimbra, Portugal: ADAI-Associacao para o Desenvolvimento da Aerodinamica Industrial, pp. 2003-2013.
- Bolourchi, P., Uysal, S., 2013, Forest fire detection in wireless sensor network using fuzzy logic, 2013 Fifth International Conference on Computational Intelligence, Communication Systems and Networks, pp. 83-87.
- Ko, B., Jung, J.H., Nam, J.Y., 2014, Fire detection and 3D surface reconstruction based on stereoscopic pictures and probabilistic fuzzy logic, Fire Safety Journal, 68, pp. 61-70.
- Olsson, H., Egberth, M., Engberg, J., Fransson, J., Pahlén, T.G., Hagner, O., Holmgren, J., Joyce, S., Magnusson, M., Nilsson, B., 2005, *Current and emerging operational uses of remote sensing in Swedish forestry*, Proceedings

of the 5th Annual Forest Inventory and Analysis Symposium, USDA Forest Service, Portland, Oregon, pp. 39-46.

- Yuan, C., Liu, Z., Zhang, Y., 2015, UAV-based forest fire detection and tracking using image processing techniques, 2015 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 639-643.
- Yuan, C., Zhang, Y., Liu, Z., 2015, A survey on technologies for automatic forest fire monitoring, detection, and fighting using unmanned aerial vehicles and remote sensing techniques, Canadian Journal of Forest Research, 45(7), pp. 783-792.
- Dampage, U., Bandaranayake, L., Wanasinghe, R., Kottahachchi, K., Jayasanka, B., 2022, Forest fire detection system using wireless sensor networks and machine learning, Scientific Reports, 12(1), 46.
- 21. Bumberger, J., Remmler, P., Hutschenreuther, T., Toepfer, H., Dietrich, P., *Potentials and limitations of wireless sensor networks for Environmental*, AGU Fall Meeting Abstracts, 2013, pp. H51F-1263.
- Chowdary, V., Kumar Gupta, M., Singh, R., 2018, A review on forest fire detection techniques: A decadal perspective, International Journal of Engineering & Technology, 7, pp. 1312-1316.
- High, D.R., Cantrell, R.L., O'Brien, J.J., 2019, System and method for operating drones under micro-weather conditions, U.S. Patent Application 16/357,485, filed October 3, 2019.
- 24. Wan, T., 2020, An investigation of quad-rotor aircraft performance under gust wind and heavy rain impacts, AIAA Scitech 2020 Forum, 1735.
- Lagkas, T., Argyriou, V., Bibi, S., Sarigiannidis, P., 2018, UAV IoT framework views and challenges: Towards protecting drones as "Things.", Sensors, 18(11), 4015.
- Mohamadi, H.E., Kara, N., Lagha, M., 2021, Efficient algorithms for decision making and coverage deployment of connected multi-low-altitude platforms, Expert Systems with Applications, 184, 115529.
- Fu, C.H., Tsao, M.W., Chi, L.P., Zhuang, Z.Y., 2021, On the dominant factors of Civilian-Use drones: A thorough study and analysis of cross-group opinions using a triple helix model (THM) with the Analytic Hierarchy Process (AHP), Drones, 5(2), 46.
- Nur, F., Alrahahleh, A., Burch, R., Babski-Reeves, K., Marufuzzaman, M., 2020, *Last mile delivery drone* selection and evaluation using the interval valued inferential fuzzy TOPSIS, Journal of Computational Design and Engineering, 7(4), pp. 397-411.
- Baek, D., Chen, Y., Bocca, A., Macii, A., Macii, E., Poncino, M., 2018, Battery-aware energy model of drone delivery tasks, Proceedings of the International Symposium on Low Power Electronics and Design, pp. 1-6.
- Rubio, A.A., Seuret, A., Ariba, Y., Mannisi, A., 2016, Optimal control strategies for load carrying drones, Delays and Networked Control Systems, pp. 183-197.
- 31. Olivares, V., Cordova, F., Sepúlveda, J.M., Derpich, I., 2015, *Modeling internal logistics by using drones on the stage of assembly of products*, Procedia Computer Science, 55, pp. 1240-1249.
- Jiao, Z., Zhang, Y., Xin, J., Mu, L., Yi, Y., Liu, H., Liu, D., 2019, A deep learning based forest fire detection approach using UAV and YOLOV3, 2019 1st International Conference on Industrial Artificial Intelligence (IAI), pp. 1-5.
- Julian, K.D., Kochenderfer, M.J., 2019, Distributed wildfire surveillance with autonomous aircraft using deep reinforcement learning, Journal of Guidance, Control, and Dynamics, 42(8), pp. 1768-1778.
- Sherstjuk, V., Zharikova, M., Sokol, I., 2018, Forest fire-fighting monitoring system based on UAV team and remote sensing, 2018 IEEE 38th International Conference on Electronics and Nanotechnology (ELNANO), pp. 663-668.
- Skorput, P., Mandzuka, S., Vojvodic, H., 2016, *The use of unmanned aerial vehicles for forest fire monitoring*, 2016 International Symposium ELMAR, pp. 93-96.
- Atanassov, K.T., Vassilev, P., Atanassova, V., Roeva, O., Iliev, R., Zoteva, D., Bureva, V., Mavrov, D., Alexandrov, A., 2021, *Generalized net model of forest zone monitoring by UAVs*, Mathematics, 9(22), 2874.
- Ghamry, K.A., Kamel, M.A., Zhang, Y., 2016, Cooperative forest monitoring and fire detection using a team of UAVs-UGVs, 2016 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1206-1211.
- Hu, J., Niu, H., Carrasco, J., Lennox, B., Arvin, F., 2022, Fault-tolerant cooperative navigation of networked UAV swarms for forest fire monitoring, Aerospace Science and Technology, 123, 107494.
- Bailon-Ruiz, R., Lacroix, S., 2020, Wildfire remote sensing with UAVs: A review from the autonomy point of view, 2020 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 412-420.
- Pamučar, D., Petrović, I., Ćirović, G., Stević, Ž., 2022, An extension of the MABAC and OS model using linguistic neutrosophic numbers: Selection of unmanned aircraft for fighting forest fires, Transport, 37(2), pp. 73-97.
- 41. Kumar, R., Kumar, D., 2016, Hybrid swarm intelligence energy efficient clustered routing algorithm for wireless sensor networks, Journal of Sensors, 2016, 5836913.
- Singh, P.K., Sharma, A., 2017, An insight to forest fire detection techniques using wireless sensor network, 2017 4th International Conference on Signal Processing, Computing and Control (ISPCC), pp. 647–653.
- 43. Mohsan, S.A.H., Khan, M.A., Noor, F., Ullah, I., Alsharif, M.H., 2022, *Towards the unmanned aerial vehicles* (UAVs): A comprehensive review, Drones, 6(6), 147.

- Stanujkic, D., Karabasevic, D., Popović, G., Pamučar, D., Stević, Ž., Zavadskas, E.K., Smarandache, F., 2021, A single-valued neutrosophic extension of the EDAS method, Axioms, 10(4), 245.
- Karasan, A., Kahraman, C., 2018, A novel interval valued neutrosophic EDAS method: prioritization of the United Nations national sustainable development goals, Soft Computing, 22(15), pp. 4891-4906.
- Simic, V., Milovanović, B., Pantelić, S., Pamučar, D., Tirkolaee, E.B., 2023, Sustainable route selection of petroleum transportation using a type-2 neutrosophic number based ITARA-EDAS model, Information Sciences, 622, pp. 732-754.
- Irvanizam, I., Syahrini, I., Zi, N.N., Azzahra, N., Iqbal, M., Marzuki, M., Subianto, M., 2021, *An improved EDAS method based on bipolar neutrosophic set and its application in group decision-making*, Applied Computational Intelligence and Soft Computing, 2021, pp. 1-16.
- Eghbali-Zarch, M., Zabihi, S.Z., Masoud, S., 2023, A novel fuzzy SECA model based on fuzzy standard deviation and correlation coefficients for resilient-sustainable supplier selection, Expert Systems with Applications, 231, 120653.
- Rekik, S., Alimi, S.E., 2023, Optimal wind-solar site selection using a GIS-AHP based approach: A case of Tunisia, Energy Conversion and Management: X, 18, 100355.
- 50. Sharaf, I.M., 2023, A new approach for spherical fuzzy TOPSIS and spherical fuzzy VIKOR applied to the evaluation of hydrogen storage systems, Soft Computing, 27(8), pp. 4403-4423.
- Banadkouki, M.R.Z., 2023, Selection of strategies to improve energy efficiency in industry: A hybrid approach using entropy weight method and fuzzy TOPSIS, Energy, 279, 128070.
- Mishra, A.R., Rani, P., Deveci, M., Gokasar, I., Pamučar, D., Govindan, K., 2023, Interval-valued Fermatean fuzzy heronian mean operator-based decision-making method for urban climate change policy for transportation activities, Engineering Applications of Artificial Intelligence, 124, 106603.
- Habib, A., Khan, Z.A., Riaz, M., Marinković, D., 2023, Performance evaluation of healthcare supply chain in Industry 4.0 with linear diophantine fuzzy sine-trigonometric aggregation operations, Mathematics, 11(12), 2611.
- Lozano, F., Jurado, J., Lozano-Galant, J., de la Fuente, A., Turmo, J., 2023, Integration of BIM and Value model for sustainability assessment for application in bridge projects, Automation in Construction, 152, 104935.
- Gopal, N., Panchal, D., 2021, A structured framework for reliability and risk evaluation in the milk process industry under fuzzy environment, Facta Universitatis-Series Mechanical Engineering, 19(2), pp. 307-333.
- Menekse, A., Ertemel, A.V., Akdag, H.C., Gorener, A., 2023, Additive manufacturing process selection for automotive industry using Pythagorean fuzzy CRITIC EDAS, PLoS ONE, 18(3), e0282676.
- Dhumras, H., Bajaj, R.K., 2023, Modified EDAS method for MCDM in robotic agrifarming with picture fuzzy soft Dombi aggregation operators, Soft Computing, 27(8), pp. 5077-5098.
- Lin, K.P., Hung, K.C., 2011, An efficient fuzzy weighted average algorithm for the military UAV selecting under group decision-making, Knowledge-Based Systems, 24(6), pp. 877-889.
- 59. Hamurcu, M., Eren, T., 2020, Selection of unmanned aerial vehicles by using multicriteria decision-making for defence, Journal of Mathematics, 2020, pp. 1-11.
- Radovanović, M., Petrovski, A., Žindrašič, V., Ranđelović, A., 2021, Application of the fuzzy AHP -VIKOR hybrid model in the selection of an unmanned aircraft for the needs of tactical units of the armed forces, Scientific Technical Review, 71(2), pp. 26-35.
- Aragão, F.V., Cavicchioli Zola, F., Nogueira Marinho, L.H., De Genaro Chiroli, D.M., Braghini Junior, A., Colmenero, J.C., 2020, *Choice of unmanned aerial vehicles for identification of mosquito breeding sites*, Geospatial Health, 15(1). pp. 92-100.
- Rakhade, R.D., Patil, N.V., Pardeshi, M.R., Mhasde, C.S., 2021. Optimal choice of agricultural drone using MADM methods. International Journal of Technological Innovation in Modern Engineering and Science (IJTIMES), 7(4), e-ISSN: 2455-2585.
- Banik, D., Ibne Hossain, N.U., Govindan, K., Nur, F., Babski-Reeves, K., 2022, A decision support model for selecting unmanned aerial vehicle for medical supplies: Context of COVID-19 pandemic, The International Journal of Logistics Management, 34(2), pp. 473-496.
- Zhang, S., Wei, G., Gao, H., Wei, C., Wei, Y., 2019, EDAS method for multiple criteria group decision making with picture fuzzy information and its application to green suppliers selections, Technological and Economic Development of Economy, 25(6), pp. 1123-1138.
- 65. Belton, V., Gear, T., 1983, On a short-coming of Saaty's method of analytic hierarchies, Omega, 11(3), pp. 228-230.
- Keshavarz-Ghorabaee, M., Amiri, M., Zavadskas, E.K., Turskis, Z., Antucheviciene, J., 2018, A comparative analysis of the rank reversal phenomenon in the EDAS and TOPSIS methods, Economic Computation & Economic Cybernetics Studies & Research, 52(3), pp. 121-134.
- 67. Li, Y., Wang, Y., Liu, P., 2016, Multiple attribute group decision-making methods based on trapezoidal fuzzy two-dimension linguistic power generalized aggregation operators, Soft Computing, 20(7), pp. 2689-2704.

26

- Zhang, H., Wang, J., Chen, X., 2014, Interval neutrosophic sets and their application in multicriteria decision making problems, The Scientific World Journal, 2014, 645953.
- Yazdani, M., Torkayesh, A.E., Stević, Ž., Chatterjee, P., Ahari, S.A., Hernandez, V.D., 2021, An interval valued neutrosophic decision-making structure for sustainable supplier selection, Expert Systems with Applications, 183, 115354.
- Ye, J., 2014, Similarity measures between interval neutrosophic sets and their applications in multicriteria decision-making, Journal of Intelligent and Fuzzy Systems, 26(1), pp. 165-172.
- Saeed, A.S., Younes, A.B., Cai, C., Cai, G., 2018, A survey of hybrid unmanned aerial vehicles, Progress in Aerospace Sciences, 98, pp. 91-105.
- 72. Stewart, M., Martin, S., Barrera, N. (Ed.), 2021, Unmanned aerial vehicles: fundamentals, components, mechanics, and regulations, Unmanned Aerial Vehicles, pp. 1-70.
- 73. Brovchenko, D.V., 2015, *Application features of UAVs different types*, 2015 IEEE International Conference Actual Problems of Unmanned Aerial Vehicles Developments (APUAVD), pp. 26-29.
- Lin, K.P., Hung, K.C., 2011, An efficient fuzzy weighted average algorithm for the military UAV selecting under group decision-making, Knowledge-Based Systems, 24(6), pp. 877-889.
- Radoglou-Grammatikis, P., Sarigiannidis, P., Lagkas, T., Moscholios, I., 2020, A compilation of UAV applications for precision agriculture, Computer Networks, 172, 107148.
- Hassanalian, M., Abdelkefi, A., 2017, Classifications, applications, and design challenges of drones: A review, Progress in Aerospace Sciences, 91, pp. 99-131.
- 77. DJL, Enterprise Drones, 2022, https://www.dji.com/ (last access: 04.04.2022).
- Petrides, P., Kyrkou, C., Kolios, P., Theocharides, T., Panayiotou, C., 2017, *Towards a holistic performance evaluation framework for drone-based object detection*, 2017 International Conference on Unmanned Aircraft Systems (ICUAS), pp. 1785-1793.
- Gao, M., Hugenholtz, C.H., Fox, T.A., Kucharczyk, M., Barchyn, T.E., Nesbit, P.R., 2021, Weather constraints on global drone flyability, Scientific Reports, 11(1), 12092.
- Sohaib Khan, M., Ali Shah, S.I., Javed, A., Mumtaz Qadri, N., Hussain, N., 2021, Drone selection using multicriteria decision-making methods, 2021 International Bhurban Conference on Applied Sciences and Technologies (IBCAST), pp. 256-270.
- 81. Lv, H., Liu, F., Yuan, N., 2021, Drone presence detection by the drone's RF communication, Journal of Physics: Conference Series, 1738(1), 012044.
- Vergouw, B., Nagel, H., Bondt, G., Custers, B., 2016, Drone technology: types, payloads, applications, frequency spectrum issues and future developments, The Future of Drone Use: Opportunities and Threats from Ethical and Legal Perspectives, 27, pp. 21-45.
- Krull, W., Tobera, R., Willms, I., Essen, H., von Wahl, N, 2012, Early forest fire detection and verification using optical smoke, gas and microwave sensors, Procedia Engineering, 45, pp. 584-594.
- Duque, L., Seo, J., Wacker, J., 2018, Synthesis of unmanned aerial vehicle applications for infrastructures, Journal of Performance of Constructed Facilities, 32(4), 04018046.
- 85. Rothstein, A., 2015, Drone, Bloomsbury Publishing USA.
- Savkin, A.V., Matveev, A.S., Hoy, M., Wang, C., 2015, Safe Robot Navigation among moving and steady obstacles, Elsevier Science.
- Wang, B.H., Wang, D.B., Ali, Z.A., Ting Ting, B., Wang, H., 2019, An overview of various kinds of wind effects on unmanned aerial vehicle, Measurement and Control, 52(7–8), pp. 731-739.
- Li, N., Liu, X., Yu, B., Li, L., Xu, J., Tan, Q., 2021, Study on the environmental adaptability of lithium-ion battery powered UAV under extreme temperature conditions, Energy, 219, 119481.
- Gorcun, O.F., Senthil, S., Kucukonder, H., 2021, Evaluation of tanker vehicle selection using a novel hybrid fuzzy MCDM technique, Decision Making, 4(2), pp. 140-162.
- Karasan, A., Bolturk, E., Kahraman, C., 2020, An integrated interval-valued neutrosophic AHP and TOPSIS methodology for sustainable cities' challenge, Advances in Intelligent Systems and Computing, 1029, pp. 653-661.
- Gulum, P., Ayyildiz, E., Taskin Gumus, A., 2021, A two level interval valued neutrosophic AHP integrated TOPSIS methodology for post-earthquake fire risk assessment: An application for Istanbul, International Journal of Disaster Risk Reduction, 61, 102330.
- Karasan, A., Bolturk, E., Kahraman, C., 2019, A novel neutrosophic CODAS method: Selection among wind energy plant locations, Journal of Intelligent and Fuzzy Systems, 36(2), pp. 1491-1504.